Author's response to reviews

Title: A novel mutation in the tyrosine kinase domain of ERBB2 in hepatocellular carcinoma

Authors:

Tanios Bekaii-Saab (Tanios.Bekaii-Saab@osumc.edu)
Nita Williams (Nita.Williams@osumc.edu)
Christoph Plass (Christoph.Plass@osumc.edu)
Miguel Villalona Calero (Miguel.Villalona@osumc.edu)
Charis Eng (engc@ccf.org)

Version: 7 Date: 6 December 2006

Author's response to reviews:

All requested changes were done as below - Thanks

Please remove the text 'title' from the title page. DONE

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. DONE

Running title - Please remove the running title from the title page. DONE

Keywords section - Please remove the keywords section from the manuscript. DONE

Introduction and Background - Please rename this 'Background'.DONE

References - References should be cited in the text using consecutive numbers in square brackets starting from [1] etc. Please ensure the type of bracket used is consistent. Please do not use superscript numbers. DONE

References - In the manuscript text, please place the reference before the full stop. DONE

Conclusions - Please include a Conclusions section after the Discussions, alternatively please rename the discussions section 'Discussions and Conclusions'. CHANGED TO Discussions and Conclusions

Competing interests - Please move this section to before the Authors' contributions. DONE

Competing interests - Please write 'The authors declare that they have no competing interests'. DONE

Please move the Acknowledgements to after the Authors' contributions. DONE

Please move the Figure legends to after the References. DONE